\begin{table}%t2
%\centering
\par
\caption{\label{lumino}Photometric data and absolute magnitudes based on the parallaxes  in {Table}~\ref{parallax}.}
\begin{tabular}{llrrrrrrrr}
\hline\hline\noalign{\smallskip}
ID &  LHS &    $V$~~      &   $R$~~      & $I$~~         & $J$  ~~  & $H$	~~  & $K_{\rm s}$	~~  & $M$($J$)~~~~~~	  & $M$($K$)~~~~~~\\
\hline \\ 
1 &1047	& 11.48\tablefootmark{1}&  10.27\tablefootmark{1}  &   8.71\tablefootmark{1}  & 7.22   & 6.71  & 6.39  & $8.49 ~  (-0.14,+0.14)        $ &$7.66 ~  (-0.14,+0.14)$ \\ 
2 &1050	& 12.61\tablefootmark{2}  &  11.46\tablefootmark{2}  &  10.04\tablefootmark{2}  & 8.62   & 8.07  & 7.81  & $8.33  ~ (-0.08, +0.09)$ &$7.52 ~ (-0.08, +0.09)$ \\ 
3 &1104	& 12.54\tablefootmark{3}  &  11.35\tablefootmark{3}  &  10.34\tablefootmark{3}  & 8.93   & 8.35  & 8.10  & $6.97  ~ (-0.12, +0.12)$ &$6.13 ~ (-0.12, +0.12)$ \\ 
4 &1475	& 10.48\tablefootmark{3}  &   9.50\tablefootmark{3}  &   8.71\tablefootmark{3}  & 7.43   & 6.80  & 6.59  & $6.31  ~ (-0.17, +0.19)$ &$5.48 ~ (-0.17, +0.19)$ \\ 
5 &228	& 15.53\tablefootmark{4}  &  14.42\tablefootmark{5}  &  13.25\tablefootmark{4}  & 12.02  & 11.52 & 11.3  & $8.89  ~ (-0.10, +0.11)$ &$8.17 ~ (-0.10, +0.11)$ \\ 
6 &1923	& 18.09\tablefootmark{4}  &  15.91\tablefootmark{5}  &  13.94\tablefootmark{4}  & 11.92  & 11.38 & 11.09 & $10.14 ~ (-0.08, +0.08)$ &$9.31 ~ (-0.08, +0.08)$ \\ 
7 &243	& 16.09\tablefootmark{6}  &  14.80\tablefootmark{6}  &  13.12\tablefootmark{6}  & 11.51  & 11.04 & 10.74 & $9.58  ~ (-0.06, +0.07)$ &$8.81 ~ (-0.06, +0.07)$ \\ 
8 &1976	& 11.38\tablefootmark{3}  &  10.31\tablefootmark{3}  &   9.41\tablefootmark{3}  & 8.06   & 7.48  & 7.24  & $6.08  ~ (-0.26, +0.29)$ &$5.26 ~ (-0.26, +0.29)$ \\ 
9 &2065	& 18.96\tablefootmark{7}  &  16.78\tablefootmark{7}  &  14.49\tablefootmark{7}  & 11.21  & 10.47 & 9.94  & $11.42 ~ (-0.09, +0.09)$ &$10.1 ~ (-0.09, +0.09)$ \\ 
10&306	& 14.19\tablefootmark{8}  &  12.80\tablefootmark{8}  &  11.05\tablefootmark{8}  & 9.36   & 8.76  & 8.50  & $9.00  ~ (-0.44, +0.56)$ &$8.14 ~ (-0.44, +0.56)$ \\ 
11&2472	& 12.81\tablefootmark{2}  &  11.61\tablefootmark{2}  &  10.12\tablefootmark{2}  & 8.70   & 8.19  & 7.87  & $8.43  ~ (-0.09, +0.09)$ &$7.6  ~ (-0.09, +0.09)$ \\ 
12&326	& 14.93\tablefootmark{9}  &  13.99\tablefootmark{9}  &  13.05\tablefootmark{9}  & 11.93  & 11.43 & 11.23 & $7.06  ~ (-0.74, +1.14)$ &$6.36 ~ (-0.74, +1.14)$ \\ 
13&334	& 18.02\tablefootmark{10}& 16.70\tablefootmark{10} &15.13\tablefootmark{10}& 13.75  & 13.25 & 13.04 & $10.47 ~ (-0.35, +0.42)$ &$9.77 ~ (-0.35, +0.42)$ \\ 
14&2632	& 19.15\tablefootmark{4}  &  16.80\tablefootmark{5}  &  14.76\tablefootmark{4}  & 12.23  & 11.58 & 11.21 & $10.86 ~ (-0.09, +0.10)$ &$9.84 ~ (-0.09, +0.10)$ \\ 
15&...   & 14.52\tablefootmark{3} &  13.35\tablefootmark{3}  &  11.88\tablefootmark{3}  & 9.48   & 8.91  & 8.61  & $9.31  ~ (-0.11, +0.12)$ &$8.44 ~ (-0.11, +0.12)$ \\ 
16&2686	& 14.16\tablefootmark{11}& 12.88\tablefootmark{11} &11.18\tablefootmark{11} & 9.58   & 9.00  & 8.69  & $9.51  ~ (-0.10, +0.11)$ &$8.62 ~ (-0.10, +0.11)$ \\ 
17&2719	&  ...       &  14.84\tablefootmark{5}  &  13.09\tablefootmark{5}  & 11.28  & 10.77 & 10.48 & $9.12  ~ (-0.19, +0.21)$ &$8.33 ~ (-0.19, +0.21)$ \\ 
18&360	& 15.22\tablefootmark{9}  &  14.29\tablefootmark{9}  &  13.41\tablefootmark{9}  & 12.39  & 11.85 & 11.62 & $8.64  ~ (-0.34, +0.40)$ &$7.87 ~ (-0.34, +0.40)$ \\ 
19&369	& 12.96\tablefootmark{2}  &  11.75\tablefootmark{2}  &  10.25\tablefootmark{2}  & 8.74   & 8.19  & 7.98  & $7.93  ~ (-0.12, +0.13)$ &$7.17 ~ (-0.12, +0.13)$ \\ 
20&2924	& 19.58\tablefootmark{12} & 17.28\tablefootmark{5}  & 15.21\tablefootmark{12} & 11.99  & 11.23 & 10.74 & $11.62 ~ (-0.14, +0.15)$ &$10.3 ~ (-0.14, +0.15)$ \\ 
21&3343	& 11.68\tablefootmark{3}  &  10.42\tablefootmark{3}  &   9.32\tablefootmark{3} & 7.85   & 7.25  & 7.00  & $7.02  ~ (-0.07, +0.07)$ &$6.17 ~ (-0.07, +0.07)$ \\ 
22&3445	& 11.59\tablefootmark{3}  &  10.31\tablefootmark{3}  &   9.16\tablefootmark{3}  & 7.58   & 7.03  & 6.81  & $6.31  ~ (-0.09, +0.09)$ &$5.54 ~ (-0.09, +0.09)$ \\ 
23&3482	&  9.79\tablefootmark{3}  &   8.71\tablefootmark{3}  &   7.82\tablefootmark{3}  & 6.88   & 6.22  & 6.04  & $6.06  ~ (-0.11, +0.11)$ &$5.22 ~ (-0.11, +0.11)$ \\ 
24&3494	& 13.41\tablefootmark{3}  &  11.41\tablefootmark{3}  &   9.76\tablefootmark{3}  & 7.79   & 7.19  & 6.85  & $9.58  ~ (-0.04, +0.04)$ &$8.64 ~ (-0.04, +0.04)$ \\ 
25&528	& 11.76\tablefootmark{3}  &  10.56\tablefootmark{3}  &   9.52\tablefootmark{3}  & 8.06   & 7.38  & 7.18  & $6.86  ~ (-0.09, +0.09)$ &$5.98 ~ (-0.09, +0.09)$ \\ 
26&3872	& 13.87\tablefootmark{3}  &  12.43\tablefootmark{3}  &  11.20\tablefootmark{3}  & 9.65   & 9.09  & 8.81  & $7.69  ~ (-0.23, +0.26)$ &$6.85 ~ (-0.23, +0.26)$ \\ 
27&535	& 12.45\tablefootmark{3}  &  11.20\tablefootmark{3}  &  10.10\tablefootmark{3}  & 8.62   & 8.10  & 7.92  & $8.05  ~ (-0.08, +0.08)$ &$7.34 ~ (-0.08, +0.08)$ \\ 
\hline                                                                                                                                 
\end{tabular}
\tablebib{\tablefoottext{1}{\citet{2007MNRAS.380.1261K};} 
    \tablefoottext{2}{\citet{1990AAS...83..357B};}  
    \tablefoottext{3}{\citet{1996AJ....112.2300W};} 
    \tablefoottext{4}{TOPP see text};
    \tablefoottext{5}{GSC2.3 transformation see text}; 
    \tablefoottext{6}{\citet{1989PASP..101..614D};} 
    \tablefoottext{7}{\citet{2005AJ....130..337C};} 
    \tablefoottext{8}{\citet{2005AJ....129.1954J};}
    \tablefoottext{9}{\citet{2006AJ....132.1234C};} 
    \tablefoottext{10}{\citet{2008AJ....136..840J};}
    \tablefoottext{11}{\citet{1979PASP...91..193W};}
    \tablefoottext{12}{\citet{2002AJ....124.1170D}}.}
\tablefoot{All infrared magnitudes come from the 2MASS catalogue. Absolute magnitude  uncertainties include both parallax and apparent magnitude errors.}
\vspace*{2mm}
\end{table}